**Author’s response to reviews**

**Title:** Trauma burden in Tanzania: A one-day survey of all district and regional public hospitals

**Authors:**

Hendry Sawe (hendry_sawe@yahoo.com)
Juma Mfinanga (jumamfinanga@gmail.com)
Khalid Mbaya (khalidrajabu@yahoo.com)
Phillip Koka (philipkoka@gmail.com)
Said Kilindimo (skilindimo@yahoo.com)
Michael Runyon (michael.runyon@carolinashealthcare.org)
Victor Mwafongo (mwafongo@muhas.ac.tz)
Lee Wallis (lee.wallis@uct.ac.za)
Teri Reynolds (reynoldst@who.int)

**Version:** 1 **Date:** 26 Jul 2017

**Author’s response to reviews:**

Hendry R. Sawe
Emergency Medicine Department
Muhimbili University of Health and Allied Sciences
P.O.Box 65014
Dar es salaam
Tanzania

Editor,

BMC Emergency Medicine
Dear Editor,

Re: Re-submission of manuscript titled “Trauma burden in Tanzania: A one-day survey of all district and regional public hospitals”

Kindly refer to the above title.

I write to submit the reviewed manuscript after receiving comments from editor and reviewers.

With this letter, I have attached the responses (point by point) to all the comments made by all the reviewers on our manuscript.

I hope that our manuscript will be considered for publication

Thank you in advance

Sincerely

Hendry R. Sawe, MD, MMED, MBA